Author's response to reviews

Title: With good intentions: complexity in unsolicited informal support for Aboriginal and Torres Strait Islander peoples. A qualitative study.

Authors:

Nathaniel J Ward (harlequindreaming@gmail.com)
Tanisha Jowsey (Tanisha.Jowsey@anu.edu.au)
Penny J Haora (Penny.Haora@anu.edu.au)
Clive Aspin (Clive.Aspin@Sydney.edu.au)
Laurann E Yen (Laurann.Yen@anu.edu.au)

Version: 4 Date: 30 August 2011

Author's response to reviews:

Dear Robyn,

My apologies- I have harmonised the author list, and corrected the two references so that they attach to the correct paper-

kind regards

Laurann Yen